Author’s response to reviews

Title: Effects of low-flow sevoflurane anesthesia on renal function in low birth weight infants

Authors:

Na Xing Dr (mzknana@126.com)
Xin Wei Dr (787611423@qq.com)
Yanzi Chang Dr (Minerva@163.com)
Yingying Du Dr (duyy87@126.com)
Wei Zhang Dr (15003995323@163.com)

Version: 8 Date: 24 December 2014

Author’s response to reviews: see over
Dear Tom Rowles PhD,

Based on reviewers' comments, we have revised the manuscript, and the modifications are shown in red font. Moreover, the yellow highlighted content show the key modifications. We would like to make the following points to clarify certain aspects of our study:

**Methods**

P 4 L 55: The first sentence has been removed from the Methods section. Our study has been registered with Chinese Clinical Trial Registry and the registration number is ChiCTR-ONRC-14004771. The important sentence that the study was registered with Chinese Clinical Trial Registry (registration no: ChiCTR-ONRC-14004771) has been added to the manuscript.

We look forward to your reply, and thank you very much for your consideration.

Yours sincerely,

Na Xing